Author's response to reviews

Title: Comprehensive profiling of DNA methylation in colorectal cancer reveals subgroups with distinct clinicopathological and molecular features

Authors:

Pei Woon Ang (angp02@student.uwa.edu.au)
Marie Loh (csimlcs@nus.edu.sg)
Natalia Liem (nmiv11@nus.edu.sg)
Pei Li Lim (peili.85@gmail.com)
Fabienne Grieu (fabiennegrieu@yahoo.com.au)
Aparna Vaithilingam (csiav@nus.edu.sg)
Cameron Platell (cameron.platell@uwa.edu.au)
Wei Peng Yong (wei_peng_yong@nuhs.edu.sg)
Barry Iacopetta (barry.iacopetta@uwa.edu.au)
Richie Soong (csirs@nus.edu.sg)

Version: 4 Date: 11 May 2010

Author's response to reviews:

Updated affiliations to the following format: Department, Institute, (Street), City, (Postal code), Country
Removed underlining from manuscript
Updated reference 25 to include full details
Changed “Figure” to “Fig.” where present
Added "in the GoldenGate Methylation Array data” to the end of paragraph 2 on page 10 to address the remaining comment from Reviewer 2.